Author’s response to reviews

Title: Respiratory compromise in children presenting to an urban emergency department of a tertiary hospital in Tanzania: A descriptive cohort study

Authors:

Biita Muhanuzi (biitamuhanuzi@yahoo.com)
Hendry Sawe (hendry_sawe@yahoo.com)
Said Kilindimo (skilindimo@yahoo.com)
Juma Mfinanga (jumamfinanga@gmail.com)
Ellen Weber (ellen.weber@me.com)

Version: 4 Date: 07 Feb 2019

Author’s response to reviews:

COMMENT

Please fix references - again, I reiterate it's unacceptable to cite Wikipedia. Most of the references are cited incorrectly still. Please don't cite UpToDate; cite the references from UpToDate directly instead. The figure still contains errors. Aside from these, thank you for considering the other comments. I think the manuscript is better shape now.

RESPONSE

Thank you very much for the comment, we have re-done all the references and fixed as recommended. Line 268-363